

Article

# MDPI

# **Executive Incentives Matter for Corporate Social Responsibility under Earnings Pressure and Institutional Investors Supervision**

## Lili Ding <sup>1,2</sup>, Zhongchao Zhao <sup>1</sup> and Lei Wang <sup>1,2,\*</sup>

- <sup>1</sup> School of Economics, Ocean University of China, Qingdao 266100, China; Ilding@ouc.edu.cn (L.D.); chaozzouc@stu.ouc.edu.cn (Z.Z.)
- <sup>2</sup> Marine Development Studies Institute of OUC, Key Research Institute of Humanities and Social Sciences at Universities, Ministry of Education, Qingdao 266100, China
- \* Correspondence: leiwang@ouc.edu.cn

Received: 4 February 2020; Accepted: 19 March 2020; Published: 22 March 2020



**Abstract:** This paper theoretically explores the impact of the incentive preferences of executives (i.e., short-term incentives and long-term incentives) on corporate social responsibility (CSR) decisions (i.e., institutional CSR and technical CSR). Further, the paper presents the mechanism through which executives influence CSR activities by the pressures from financial analysts and institutional investors supervision. Using a large sample of China-listed firms over 2007–2017, we achieve some helpful empirical results. The executives with short-term incentives tend to implement technical CSR strategy, while those with long-term incentives tend to implement institutional investors short-term incentives, compared with those with long-term incentives, show stronger inter-temporal tradeoffs behaviors in the earnings pressure context. Furthermore, dedicated institutional investors can effectively attenuate the hypocritical behaviors of executives, and the effectiveness of governance shows a positive relationship with investors' horizon. Our findings enrich the understanding on the relationship between the executives and CSR decisions in the earnings pressure context and further helps to perfect the institutional design in China's listed companies.

**Keywords:** Corporate Social Responsibility; Earnings Pressure; Executive Incentives; Institutional Investor; Emerging Market

### 1. Introduction

On 19 August 2019, Business Roundtable announced a new statement signed by 181 CEOs who commit to lead their companies for the benefit of all stakeholders—customers, employees, suppliers, communities, and shareholders. (Since 1978, Business Roundtable has periodically issued Principles of Corporate Governance). The new statement focuses on creating long-term value for all stakeholders and prioritizing corporate social responsibility (CSR), which will result in shared prosperity and sustainability for all society. In China, as an emerging market, the Shenzhen Stock Exchange initially issued guidelines on the social responsibility of listed companies in 2006. Then, the relevant monitoring and evaluating clauses were presented to boost CSR reporting to balance the relationship between the extensive economic growth and environmental destruction [1]. More and more listed firms have taken CSR programs to demonstrate their good corporate citizenship [2]. However, many executives of China's listed firms are also severely lacking in their knowledge reserves related to the sustainable development strategy for society. Sometimes, CSR-related activities may be utilized to serve the interests of executives themselves. Hence, a more integrative framework needs to be proposed to examine CSR issues and to investigate these issues in the context of emerging economies.

The current literature focuses on the motivation of the CSR strategy implemented by executives from the two following competitive points [3,4]. First, for enhancing the harmonious relationship with the community and natural ecology, CSR activities can make the interests of executives, shareholders, and other non-financial stakeholders better aligned with firms [5]. Porter et al. [6] claim that "corporate success and social welfare are not a zero-sum game". Second, CSR activities are a form of wasteful spending, with the primary goal of enhancing the manager's private benefits at the expense of shareholders [7,8]. For example, insider-initiated corporate philanthropy is often engaged in by the top managers' own desires, not by the stakeholders' willingness.

However, the above literature neglects the analysis of why the executives carry out CSR activities or not and what motives executives' CSR decisions. Although some studies have explored the executives' CSR decisions from the views of education [9], age and tenure [10], and organization identification [11], these are only the personal features of executives. In fact, the executives' strategic decision, e.g., CSR decision, is a kind of interactive behavioral decision mostly influenced by external pressures. First, some studies show that the executives' decision can be affected by earnings pressure from financial analysts. The reason is that the investors believe in the prospect information of publicly listed firms from financial analysts, who are wise and knowledgeable experts [12]. Thus, executives of listed firms pay more attention to the financial analysts' earnings forecasts [13,14]. In order to attract the analysts' attention, executives usually perform inter-temporal behaviors to maintain high yield and share price of firms. This kind of purposeful intervention may limit investment in CSR-related activities, which is contradictory to the sustainability prospect of firms. Furthermore, the executives' decision also can be affected by the pressure from institutional investors' supervision. Since the executives cannot make decisions arbitrarily, they are constrained by the different institutional ownership structures [15]. It shows that the inter-temporal behaviors of executives may be restrained by the supervision of institutional investors [16]. These findings are conducive to understanding executives' actions on the strategic decision and inter-temporal tradeoffs. However, the existing researches pay less attention to the pressures of financial analysts and institutional investors on CSR decision by the executives. In 2019, Qian et al. [17] presented a helpful attempt to examine the impact of financial analysts on a firm's corporate social performance. They neglect the impact of the internal incentives of executives on CSR decisions. Motivation theory has proved that internal incentives, such as salary, bonus, and stock options, deeply affect executives' behavior. Hence, it is essential to construct an integrative framework, which can explain the relationship between the incentive preference of executives and CSR strategy when considering the pressures from financial analysts and institutional investors' supervision.

The main purpose of this paper is to provide an analyzing framework to conceptualize the impact of the incentive preferences of executives (i.e., short-term incentives and long-term incentives) on CSR decisions (i.e., institutional CSR and technical CSR) under the pressures from financial analysts and institutional investors supervision. We extend the work of Qian et al. [17] by introducing internal incentives and external pressure situations. We use a sample of China A-share–listed firms over 2007–2017 to study the issues discussed above, which would help to discover the status of China's corporate governance.

The contributions of this paper are presented as follows. First, this paper classifies the incentive preferences of executives as short-term incentives and long-term incentives. We investigate the relationship between these two different preferences of executives and the CSR strategy decision (e.g., institutional CSR and technical CSR). Our findings show that executives with long-term incentives prefer to choose institutional CSR items to enhance the harmonious relationship with the community, while those with short-term incentives tend to choose technical CSR items. Second, this paper attempts to incorporate the pressure from financial analysts into the theoretical framework explaining the relationship between the incentive preference of executives and CSR strategy decisions. Our findings can explain the reason why executives with short-incentives perform inter-temporal tradeoffs for short-term financial performance in the context of earnings pressure. Third, the pressures from institutional investors are divided into four distinct groups by portfolio concentration and investment

horizon. They are integrated into the theoretical framework proposed above. The finding can present supervision effects from institutional investors when the executives with different incentive preferences make CSR decisions.

The remainder of this article is organized as follows. Section 2 reviews the relevant literature of this study and constructs a theoretical framework. Section 3 describes the data. Section 4 provides descriptive analysis and empirical results and briefly describes robustness tests to enhance the reliability of the conclusion. Finally, Section 5 concludes.

#### 2. Literature Review, Research Hypothesis, and Theoretical Framework

#### 2.1. The Executive Incentives and Corporate Social Responsibility Strategy

According to the above literature, executives are essential participants of CSR decisions. In this paper, we investigate the preferences of executives, which can influence executives' CSR decisions. The executive incentives show a significant impact on the strategic decision of firms, including the CSR strategy. According to Finkelstein and Hambrick [18], executive incentives provide an effective way to measure the convergence of the interests of executives and shareholders. Specifically, compared with executives with short-incentives, these with long-incentives are more likely to converge with the interests of shareholders and pay more attention to capital investment [19], and are less sensitive to short-term financial performance [20]. Some evidence also supports that executive incentives could affect the CSR strategy. For example, Fabrizi et al. [21] find that both monetary and non-monetary incentives have an effect on CSR decisions. Hong and Minor [22] find that executives with direct incentives for CSR strategy are a useful tool to increase firm social performance.

According to the research of Souder and Bromiley [23], the firm s operating performance related to executive incentives can result in the agent problem because of incentive preferences mismatch. For example, the executives with short-incentives allocate more resources to projects and schemes that promote short-term performance and serve the interests of executives. These activities are not beneficial to the sustainable development of firms. Oh et al. [24] point out that older CEOs may be less likely to perform long-term investments, e.g., CSR strategy, and more likely take short-term profit-generating initiatives. Antia et al. [25] also show that short-term executive incentives are an essential indicator of investment to chase faster returns. Marinovic and Varas [20] note that executives who are more sensitive to short-term performance are more likely to engage in performance manipulation.

Therefore, we need to classify executive incentives when implementing CSR strategy. The executives with short-term incentives incline to curtail the use of CSR strategy to promote financial performance in the short term due to their career security and reputation. In contrast, executives with long-term incentives pay more attention to the sustainable development of the firm and are more likely to implement CSR strategy because CSR-related activities could promote the long-term value and sustainable development of firms in the long run.

Based on the discussion above, this paper proposes the following hypothesis:

#### **Hypothesis 1a.** The short-term incentives of executives show a negative relationship with CSR strategy.

### Hypothesis 1b. The long-term incentives of executives show a positive relationship with CSR strategy.

#### 2.2. The Executive Incentives and CSR Strategy when Considering Earnings Pressure from Financial Analysts

Earnings pressure may result in an earnings expectation gap between financial analysts and executives [26]. Some of the literature supports the idea that earnings pressure stimulates myopic behavior in executives, such as earnings management [24], and increases the current earnings at the expense of future earnings [17]. These executives attempt to weaken earnings pressure by cutting strategic investments [27,28] and corporate downsizing [29]. Duong and Pescetto [30] also show that

earning management driven by earnings pressure may be implemented to maintain high duration valuations of the market.

In addition, some studies show that the reaction of executives to earnings pressure depends on their different characteristics and equity institutions. For example, compared with executives with short-term incentives, these with long-term incentives are less responsive to earnings pressure [15]. Lee and Chang [31] point out that executive incentives are likely to affect the reactions of executives to earnings pressure. Moreover, ownership structure shows a significant impact on the response of executives. Schulz and Wiersema [29] show that institutional investor stock ownership weakens executives' sensitivity under market pressure. Generally, the earnings pressure plays an essential role in investigating the impact of executive incentives on CSR strategy.

Based on the discussion above, this paper puts the following hypothesis:

**Hypothesis 2a.** The short-term incentives of executives show a more negative relationship with CSR strategy under earnings pressure.

**Hypothesis 2b.** *The long-term incentives of executives show an insignificant relationship with CSR strategy under earnings pressure.* 

#### 2.3. The Executive Incentives and CSR Strategy when Considering Pressure from Institutional Investors Supervision

As an effective monitoring power in emerging markets, institutional investors play an important role in restraining executive short-termism and myopic behaviors [32]. According to the research of Gillan and Starks [33] and Connelly et al. [16], they point out that interest convergence between executives and shareholders can be gained through multiple governance mechanisms. For example, Alvarez et al. [34] find that institutional investors can influence the investment decisions of executives. The institutional shareholders can promote the implementation of long-term strategy by executives and CSR practices of enterprises, improving investor wealth [35]. Moreover, Fich et al. [36] show that the monitoring attention of institutional investors depends on portfolio positions. Dedicated investors have more incentives to monitor the behaviors of executives and show greater tolerance for short-term performance disappointment [37]. Inversely, transient investors frequently trade in and out of firms on the base of change in stock market value and have fewer incentives to behaviors of executives [38]. Li et al. [39] provide evidence that institutional shareholders with short-term horizon are more likely to loosen regulatory constraints. Inversely, institutional shareholders with long-term horizon show an essential supervision function in restraining myopic managerial behaviors [40]. However, there are few studies that explain the relationship between institutional shareholders with short-term horizon when considering the governance function of institutional investors. Therefore, this paper provides interesting governance hypotheses considering the portfolio and investment horizon of institutional investors, which can help to clarify the governance mode of distinct classification groups.

Based on the discussion above, this paper proposes the following hypothesis:

**Hypothesis 3a.** The short-term incentives of executives show an insignificant relationship with CSR strategy under dedicated institutional investors with long-term horizon

**Hypothesis 3b.** *The long-term incentives of executives show a more positive relationship with CSR strategy under dedicated institutional investors with long-term horizon* 

**Hypothesis 3c.** The short-term incentives of executives show a more negative relationship with CSR strategy under transient institutional investors with short-term horizon

**Hypothesis 3d.** *The long-term incentives of executives show an insignificant relationship with CSR strategy under transient institutional investors with short-term horizon* 

#### 2.4. Theoretical Framework

To investigate the complex relationship between the incentive preference of executives and CSR strategy under the scenario of the earnings pressure and institutional investors, this paper constructs the theoretical framework, as shown in Figure 1. In this theoretical framework, the research process can be divided into three stages.



**Figure 1.** Theoretical framework for the corporate social responsibility (CSR) preference of executives under earnings pressure.

#### 3. Data

#### 3.1. Corporate Social Responsibility

The measurements of CSR performance in this study are based on the Chinese Research Data Services Platform (abbr. CNRDS), providing the CCSR database. (The databases providing the measurement of CSR performance of China listed firms are the CNRDS, RKS, and HEXUN Web CSR rating dada in China. This paper takes CNRDS as the primary database to measure the CSR scores because the CSR scores from CNRDS and RKS show significant consistency, and CNRDS could provide more qualitative indicators than RKS). The CCSR database takes the KLD STATA database as the mainframe model and provides various indicators of CSR measurement based on the disclosed annual CSR reports by the Shanghai Stock Exchange (SSE) and Shenzhen Stock Exchange (SZSE) since 2006. CCSR evaluates CSR performance of Chinese listed firms based on six qualitative indicators (CCSR provides the six qualitative indicators about CSR performance, which are the community, corporate governance, diversity, employee relations, environment, and product quality and safety, respectively) and statistics annually from two dimensions of strengths and concerns.

This paper divides the CSR-related activities into institutional CSR and technical CSR (i.e., ICSR and TCSR) according to the method of Freeman et al. [41]. The former refers to some responsible activities serving the outside stakeholders of enterprises, such as community relations, environmental stewardship, and diversity. (Institutional CSR is more likely to generate moral capital, reduce the enterprise's financing constraints, and effectively buffer or minimize the damage to the corporate image from negative exogenous events in theory [10]). The latter refers to some responsible activities serving the inner stakeholders of enterprises, such as employee relations, product safety, and governance. (The technical CSR is more likely to generate less moral capital, which may embed the interests of the executives into the inner CSR activities [42,43]). Following Mattingly and Berman [9] and Godfrey et al. [44], this paper defines INCSRS and INCSRNC as the sum scores of the strengths and concerns items under the institutional CSR category, and TECSRS and TECSRC as the sum scores of the strengths and concerns items under the technical CSR category, respectively. Besides, this paper calculates the net CCSR scores to measure ICSR and TCSR level, denoted by NEINCSR and NETECSR, respectively.

Based on the studies of Lee et al., [42] and Cao et al., [45], this paper classifies the executive incentives as the short-term incentives and the long-term incentives. Short-term incentives for executives (SHOEXE) are measured by the natural logarithm of the sum of salary, bonus, and other cash payments of the top three senior executives. When the financial performance of enterprises is better at the end of the year, the executives will obtain more in terms of salary, bonus, and other cash payments. Therefore, executives with relatively short-term incentives are more sensitive to short-term financial performance and thus pay more attention to short-term earnings rather than long-term performance.

Moreover, long-term incentives for executives (LONEXE) are measured by the natural logarithm of the market value of the shares and the unexercised stock options of the top three senior executives. Therefore, executives with relatively long-term incentives pay more attention to impression management and are more likely to implement CSR strategy and thus are inclined to promote future financial performance.

#### 3.3. Earnings Pressure

This paper adopts the method of Zhang and Gimeno [26] to calculate the proxy of earnings pressure (EARPRE). Contrasted with other measures of earnings pressure, this method provides the reliability where this variable could not be affected by the endogenous events, such as earning management. Earnings pressure could be measured for firm i at the beginning of year t as

where *Analyst consensus forecast* denotes the mean of the earnings forecasts from financial analysts at the beginning of reporter term [46] and *Potential earnings* denotes the potential performance of firms at the beginning of reporter term [47]. Zhang and Gimeno [26] point out that the potential performance of executives could be influenced by the historical performance of firms and industry performance over the same period. Thus, potential earnings are calculated by historical performance containing information of stock price before the analysts' forecasts, and the average peer potential earnings performance. (When calculating the variable of potential performance, this paper excludes the focal firm, and controls the same earnings benchmark at the same industry for the accounting year). This paper sets the weights of each part above to 0.6 and 0.4, respectively. (We also set the weight of historical earnings and peer earnings to 0.5, respectively, and the results are robustness).

Table 1 shows, the mean differences between analyst forecast consensus and reported earnings are significantly higher than that between potential earnings and reported earnings (paired t-test: p < 0.01). In comparison, the estimation of potential earnings is unbiased (p > 0.1). The distribution of analyst forecast consensus at the beginning of the reporter term is skewed toward overestimating reported earnings (with a mean difference significantly above 0). The results shown in Table 1 are consistent with previous literature. Thus, analyst forecasts are valuable in our study, even if they appear to be biased.

#### 3.4. Institutional Investors

Following the approach by Fich et al. [36], this paper captures the relative importance of the target firm concerned by the institutional investors. To estimate the market value owned by the institutional investors, the calculation is that the firm's market values multiply the percentage of the firms' equity held by the institutional investors. Then, we calculate the rate of the market value held by the institutional investor in a single firm to the value of institutional investor's portfolios. At last, we define dedicated institutions (DEDIINS) as those whose holding value in the target firm is in the top 10% of their portfolios, and transient institutions (TRANINS), otherwise.

In addition, this paper also calculates the proxy of the investment horizon. Since the proxy of the investment horizon cannot be directly observed, it is measured by constructing the indicator about the frequency of investor trades. According to the research of Gaspar et al. [48], the main proxy for the

investment horizon is the average holding time, which is equal to the value of the internal term divided by investment turnover where the internal term is 1. (See Appendix B) The lower investment turnover thus shows that the firm is held by investors with long-term horizon (LONINS). Similarly, the higher investment turnover thus shows that the firm is held by investors with short-term horizon (SHOINS).

This paper constructs four institutional investor variables depending on the median values of the portfolio concentration and investment horizon. They are dedicated investors with long-term horizon (LONDEDI), dedicated investors with short-term horizon (SHODEDI), transient investors with long-term horizon (LONTRAN), and transient investors with short-term horizon (SHOTRAN).

Variables	Deviation 1, for Potential Earnings Estimate <sup>I</sup>	Deviation 2, for Analysts Forecast Consensus <sup>I</sup>	
Distribution			
1 <sup>st</sup> percentile	-1.275	-0.481	
25 <sup>th</sup> percentile	-0.135	-0.006	
Median	-0.003	0.058	
75 <sup>th</sup> percentile	0.119	0.169	
99 <sup>th</sup> percentile	1.352	1.326	
Std. Dev	0.498	0.325	
Hypotheses tested $^{\mathrm{II}}$	H0: Mean (Deviation 1) = $0$	H0: Mean (Deviation 2) = $0$	H0: Mean (Deviation 1) = Mean (Deviation 2)
Mean	-0.004	0.115	-0.120
S.E.	0.004	0.003	0.004
t	-0.982	40.060 ***	-25.009 ***

Table 1. Comparison of mean differences between focal estimates and reported earnings.

**Notes:** Following Zhang and Gimeno [26], we design the experiment of the validation of the earnings pressure measure in our study). <sup>1</sup> Both deviation 1 and 2 were the focal estimate minus reported earnings. N = 10532 (firm-levels). <sup>II</sup> The first two tests are one-sample *t-tests*; the third is a two-sample *t-test*. \*\*\* p< 0.01 One-sided tests.

#### 4. Discussion

#### 4.1. Descriptive Statistics

Table 2 reports the summary statistics about the variables used in the paper (in addition to the CCSR, CMSAR, and WIND database, several other databases are used in this study. We also use the CCSR economic database and the RESET financial database to fill in the missing values of the firm-level variables), including CSR scores, executive incentives, earnings pressure, and other several firm characteristics. All non-dummy variables are winsorized at the first and 99th percentiles, and Appendix A explains their constructions.

In particular, the average NETINCSR and NETECSR scores are 6.147 and 10.841. After distinguishing these scores into strengths and concerns item scores, the average scores of INCSRS and TECSRS are 3.305 and 10.893, while the average scores of INCSRC and TECSRC are 0.239 and 0.127, respectively. (Considering the measure differences in dimensions of CSR performance, the CSR-related measures were placed in standardized processing to gain a dimensionless performance variable). Furthermore, the average scores of LONEXE and SHOEXE are 4.827 and14.007, indicating the motivation level of the long-term incentives and the short-term incentives of executives. Together, the average share is 4.066% (6.788%) in institutional ownership by DEDIINS (TRANINS) in our sample, which includes 3.677% of the total shares held by LONDEDI (3.086% held by LONDEDI) and 2.119% held by SHODEDI (1.785% held by SHODEDI). Similarly, the average shares in institutional ownership by LONINS (SHOINS) are 3.928% and 1.879%.

#### 4.2. Do Executives with Disparate Incentives Show Different CSR Types?

4.2.1. Main Results on Executive Incentives and CSR Types

In order to test H1a and H1b, we construct the following model:

$$XCSR\_Score_{i,t+1} = \beta_0 + \beta_1 mana\_incentives_{i,t} + \lambda X_{i,t} + \eta_i + \varphi_{t+1} + \xi_{i,t+1}$$
(2)

where *XCSR\_Score* denotes the CSR scores of firms, and *mana\_incentives* denotes the incentives of executives. X includes other control variables for several firm characteristics. Meanwhile, year fixed effects ( $\phi$ ) and industry fixed effects ( $\eta$ , CSRC industry code) are controlled in the model. What is more, the time interval between CSR strategy policy and executive incentives, independent variables with a lag of one period, are introduced into the model.

Variable Abbreviation	Mean	Std. Dev	St10	Median	St90	Obs
INCSRS	3.305	2.922	2	5	11	5063
INCSRC	0.239	0.426	0	0	1	5063
NEINCSR	6.147	3.097	2	6	10	5063
TECSRS	10.893	3.156	6	11	16	5063
TECSRC	0.127	0.334	0	0	1	5063
NETECSR	10.841	3.418	6	11	15	5063
LONEXE	4.827	5.217	0	0	12.814	17,305
SHOEXE	14.007	0.714	12.816	14.045	15.098	17,436
EARPRE	0.107	0.161	-0.108	0.073	0.418	10,045
DEDIINS	4.066	5.236	0.114	2.233	10.150	902
TRANINS	6.788	11.069	0.059	2.310	19.393	14,411
LONINS	3.928	4.772	0.042	1.716	14.520	13,659
SHOINS	1.879	2.506	0.018	0.619	7.615	12,550
LONDEDI	3.677	3.161	0.300	2.667	9.874	517
SHODEDI	2.119	2.413	0.044	0.980	7.216	596
LONTRAN	3.806	4.593	0.042	1.692	14.000	13,658
SHOTRAN	1.785	2.365	0.018	0.604	7.202	12,458
CASHRA	0.405	1.306	0.102	0.253	0.545	19,028
DUTYMER	1.791	0.469	1	2	2	17,436
SEPERATE	4.785	6.876	0	0	18.595	18,413
BALANCE	0.699	0.569	0.126	0.539	1.478	18,417
ROAPER	0.035	0.034	-0.016	0.029	0.097	19,346
BOMARK	1.051	0.758	0.251	0.794	2.610	18,538
INDEPER	0.367	0.055	0.333	0.333	0.429	17,312
EQUNAT	1.422	0.667	1	1	2	18,417

Table 2. Descriptive statistics.

Notes: All variables in the sample are winsorized at 1% and 99% with the period from 2007 to 2017.

Table 3 presents the results of executives with different incentives on different CSR activities. Columns (1)–(2) and (5)–(6) in Table 3 show that executives with short-term incentives (SHOEXE) and executives with long-term incentives (LONEXE) are significantly positively associated with the CSR strengths items. Besides, the coefficients of CSR scores on the executive with short-term incentives are higher than the coefficients of executives with long-term incentives. Furthermore, the coefficients of TECSRS on executives with short-term incentives are higher than the coefficients of INCSRS on executives with short-term incentives. In contrast, the coefficients are reversed for executives with long-term incentives. For example, in column (1), the coefficient on SHOEXE is 1.626, which is less than the coefficient on SHOEXE in column (5) and is significant at the 1% level, with a t-statistic of 13.36.

Columns (3)–(4) and (7)–(8) in Table 3 show that executives with short-term incentives and executives with long-term incentives are significantly positively associated with the CSR concerns items. Both executives with short-term incentives and executives with long-term incentives are significantly positively associated with CSR strengths and concerns items, but the coefficient on CSR strengths is significantly higher than that on CSR concerns. The results show that executives are more likely to enhance the sustainable development of the enterprises, rather than hinder through CSR-related activities despite the CSR types despite depending on the executive incentives, which are consistent with Liu [49]. Our analysis reveals the fact that executives with short-term incentives are

more inclined to choose technical CSR. The technical CSR could improve the company's internal governance, increasing firms' profit-making interest. This finding is opposite to the research of Souder and Bromiley [23]. This finding provides a reasonable explanation for the mixed results on the relationship between the executives and CSR decisions in previous literature. However, executives with long-term incentives prefer to choose institutional CSR to enhance the community relationship, which is consistent with Sanders [50].

Interestingly, this study shows that executives with short-term incentives preferred to implement CSR strategies than those with long-term incentives. The results are seemingly paradoxical to the literature regarding myopic executive, stating that short-incentives executives prefer to boost short-term earnings by reducing capital investment [51]. Particularly, this study empirically shows that short-term incentives executives have a strong preference for CSR strategies, which promotes the sustainable development of enterprises, not only for technical CSR but also for institutional CSR. The results demonstrate that the visibly ethical behaviors of insider-initiated CSR-related activities in China could be regarded as a particular case of manager building empires [52], an operation distress insurance [53] and corporate hypocrisy with self-interest embedded in benevolence [54].

		Dependent Variable						
	INC	SRS	INC	SRC	TEC	SRS	TEC	SRC
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
SHOFYE	1.626 ***		0.197 ***		2.055 ***		0.099 ***	
SHOLAL	(13.36)		(9.73)		(15.84)		(6.93)	
LONEYE		0.096 ***		0.010 ***		0.089 ***		0.005 ***
LONEXE		(6.56)		(4.18)		(5.65)		(2.79)
CASHRA	0.124	0.170	0.019	0.025	0.161	0.213	0.027 *	0.029 *
CASIIKA	(0.93)	(1.25)	(0.86)	(1.09)	(1.13)	(1.45)	(1.71)	(1.87)
DUTVMED	0.084	0.084	-0.028	-0.029	-0.072	-0.078	0.004	0.004
DUTIMER	(0.66)	(0.65)	(-1.34)	(-1.34)	(-0.53)	(-0.56)	(0.27)	(0.25)
CEDAD ATE	0.003	0.006	0.002	0.002	-0.010	-0.006	0.001	0.001
SEFARATE	(0.23)	(0.50)	(0.76)	(0.94)	(-0.77)	(-0.47)	(0.51)	(0.64)
DALANCE	0.826 ***	1.017 ***	0.097 ***	0.120 ***	0.794 ***	1.048 ***	0.064 ***	0.076 ***
DALANCE	(5.50)	(6.70)	(3.87)	(4.80)	(4.96)	(6.41)	(3.63)	(4.32)
POADED	-3.963	-3.069 ***	-0.542 ***	-0.435 ***	-6.036 ***	-4.927 ***	-0.316 ***	-0.262 **
KOATEK	(-4.25)	(-3.24)	(-3.50)	(-2.79)	(-6.08)	(-4.84)	(-2.89)	(-2.39)
ROMARK	0.619 ***	0.692 ***	0.091 ***	0.101 ***	0.691 ***	0.793 ***	0.064 ***	0.070 ***
DOWIARK	(6.66)	(7.34)	(5.90)	(6.46)	(6.89)	(7.81)	(6.00)	(6.42)
INDEDDED	-1.073	-0.276	-0.076	0.022	-0.526	0.503	-0.026	0.023
INDERFER	(-1.00)	(-0.25)	(-0.42)	(0.12)	(-0.46)	(0.43)	(-0.21)	(0.19)
FOUINAT	0.095	0.094	0.003	0.002	-0.041	-0.050	-0.009	-0.009
LQUINAI	(0.48)	(0.47)	(0.08)	(0.07)	(-0.19)	(-0.23)	(-0.40)	(-0.40)
CONSTANT	-18.223 ***	4.041 ***	-2.674 ***	0.031	-19.544 ***	8.771 ***	-1.420 ***	-0.061
CONSTANT	(-10.13)	(6.88)	(-8.94)	(0.32)	(-10.19)	(13.87)	(0.211)	(-0.90)
Fixed Effects	Y	Y	Y	Y	Y	Y	Y	Y
Observations	4728	4727	4728	4727	4728	4727	4728	4727
R <sup>2</sup> -adjusted	0.163	0.080	0.135	0.069	0.149	0.066	0.119	0.079
F-test	5.57 ***	5.79 ***	3.91 ***	4.09 ***	5.32 ***	5.40 ***	3.36 ***	3.54 ***

Table 3. Executive incentives and CSR types.

**Notes:** All models include year and industry fixed effects. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

#### 4.2.2. Additional Hypocrisy Evidence on CSR Types

In this section, we further test the hypothesis that executives with short-term incentives may exacerbate hypocrisy and self-interest myopic, compared with executives with long-term incentives executives. We isolate the effect of different incentives for executives on CSR types by distinguishing earnings management, taking the absolute level of discretionary accruals as the sorting criterion. This study uses a modified version of the Jones model [55,56] to measure the proxy of earnings. (See Appendix D).

Table 4 presents the results of executives with different incentives on two types of CSR by distinguishing earnings management. In columns (1) to (8) of Table 4A, the coefficients reveal that executives with short-term incentives could have a more significant positive effect on CSR strengths items, including INCSRS and TECSRS under earnings management. For example, the coefficient is 1.342 in column (1) with a t-statistic of 4.35 and 1% statistical significance, which is less than the coefficient in column (2). On the contrary, executives with short-term incentives could have a less significant positive effect on CSR strengths items, including INCSRC and TECSRC under earnings management. For example, the coefficient is 0.239 in column (3) with a t-statistic of 4.35 and 1% statistical significance, which is more than the coefficient in column (2). The coefficients show that executives with short-term incentives could disguise earnings management through CSR-related activities and embed self-interest into charity behaviors. The conclusions provide the evidence that CSR may be a hypocritical tool for executives to achieve more private gains at the expense of stakeholders, which is consistent with Arli et al. [57].

Similarly, Columns (1) to (8) of Table 4B show the consequences of executives with long-term incentives on the CSR types. Although executives with long-term incentives may conduct earnings management through CSR-related activities, their behavioral intensity is much less than that of executives with long-term incentives. This result provides evidence that CSR-related activities may be a hypocritical tool to be used to practice earnings management by executives for their reputation. This finding is not consistent with the research of Philippe and Koehler [58], who support that executives may blindly pursue financial performance and ignore ethical issues when pursuing short-term performance. Since in China, the life cycles of listed companies are shorter than European companies. In this regard, CSR-related activities are more likely to be used as a tool for serving the interest of executives and the short-term performance of the firm rather than the sustainable development of the firm.

	A: Executives with short-term incentives and empirical evidence for short-termism.								
				Depender	nt Variable				
	INCSRS		INCSRC		TEC	TECSRS		TECSRC	
	EM Low (1)	EM High (2)	EM Low (3)	EM High (4)	EM Low (5)	EM High (6)	EM Low (7)	EM High (8)	
SHOEXE	1.342 *** (4.35)	1.54 *** (5.66)	0.239 *** (4.73)	0.196 *** (4.36)	1.548 *** (4.75)	2.329 *** (8.12)	0.098 *** (2.88)	0.083 *** (2.64)	
Control vars	Y	Y	Y	Y	Y	Y	Y	Y	
<b>Fixed Effects</b>	Y	Y	Y	Y	Y	Y	Y	Y	
Observations	1127	1522	1127	1522	1127	1522	1127	1522	
R <sup>2</sup> -adjusted	0.066	0.072	0.067	0.034	0.056	0.107	0.058	0.044	
F-test	2.59 ***	2.68 ***	1.67 ***	2.29 ***	2.46 ***	2.63 ***	1.47 ***	1.89 ***	

Table 4. Executives with different incentives and empirical evidence for hypocrisy on CSR.

B: Executives with long-term incentives and empirical evidence for short-termism.

		Dependent Variable							
	INC	INCSRS		INCSRC		TECSRS		TECSRC	
	EM Low (1)	EM High (2)	EM Low (3)	EM High (4)	EM Low (5)	EM High (6)	EM Low (7)	EM High (8)	
LONEXE	0.074 ** (2.24)	0.122 *** (3.75)	0.011 ** (2.07)	0.009 ** (3.75)	0.102 *** (2.90)	0.126 *** (3.65)	0.002 (0.58)	0.008 ** (2.20)	
Control vars	Y	Y	Y	Y	Y	Y	Y	Y	
<b>Fixed Effects</b>	Y	Y	Y	Y	Y	Y	Y	Y	
Observations	1127	1521	1127	1521	1127	1521	1127	1521	
R <sup>2</sup> -adjusted	0.017	0.044	0.020	0.024	0.010	0.013	0.021	0.014	
F-test	2.62 ***	1.89 ***	1.68 ***	2.90 ***	2.51 ***	2.70 ***	1.51 ***	1.96 ***	

**Notes:** All models include year and industry fixed effects. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

# 4.3. Do Executives with Different Incentives Practice the Inter-Temporal Tradeoffs for CSR Activities under *Earnings Pressure?*

#### 4.3.1. Main Results on Executives with Different Incentives and Inter-Temporal Tradeoffs

This section investigates whether executives with disparate incentives could practice the inter-temporal tradeoffs for CSR-related activities under earnings pressure. In Table 5, the coefficients show that executives with short-term incentives have intensely motivated to practice inter-temporal tradeoffs under earnings pressure. This result is consistent with Qian et al. [17], who point out that earnings pressure stimulates the myopic behavior of executives and thus increases current earnings at the expense of future earnings. The inter-temporal tradeoffs are mainly reflected in technical CSR items, while the institutional CSR items are not distinct. For example, the interaction terms between executives with short-term incentives (SHOEXE) and earnings pressure (EARPRE) are significantly negative with technical CSR strengths items. The coefficient of interaction terms is -1.597 with a t-statistics of -2.07 and 5% statistical significance in column (5), while the coefficient of interaction may be that institutional CSR-related activities could enhance the accumulation of moral capital, and attenuate the negative effect harmful to the enterprises.

				Dependen	t Variable			
	INC	SRS	INC	SRC	TEC	SRS	TECS	SRC
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
SHOEXE	1.675 *** (8.52)		0.265 *** (7.80)		2.205 *** (10.64)		0.136 *** (5.55)	
SHOEXE ×EARPRE	-1.073 (-1.47)		-0.271 ** (-2.14)		-1.597 ** (-2.07)		-0.194 ** (-2.13)	
LONEXE		0.110 *** (5.17)		0.015 *** (4.05)		0.075 *** (3.30)		0.006 ** (2.45)
LONEXE ×EARPRE		-0.136 * (-1.69)		-0.014 (-1.00)		-0.158 * (-1.83)		-0.018 * (-1.81)
EARPRE	0.523 (0.98)	0.202 (0.46)	0.014 (0.15)	-0.086 (-1.15)	0.347 (0.62)	-0.105 (-0.23)	0.053 (-0.80)	-0.117 ** (-2.17)
CASHRA	0.462 (1.28)	0.489 (1.34)	0.105 * (1.69)	0.109 * (1.72)	0.311 (0.82)	0.328 (0.84)	0.008 (0.18)	0.010 (0.22)
DUTYMER	0.263 (1.43)	0.261 (1.40)	-0.025 (-0.79)	-0.026 (-0.81)	-0.041 (-0.21)	-0.052 (-0.26)	-0.006 (-0.28)	-0.007 (-0.28)
SEPARATE	-0.004 (-0.21)	-0.003 (-0.18)	0.000 (0.09)	0.000 (0.10)	-0.015 (-0.81)	-0.014 (-0.73)	0.000 (0.40)	0.010 (0.42)
BALANCE	0.552 ** (2.29)	0.710 *** (2.93)	0.110 *** (2.63)	0.137 *** (3.25)	0.655 *** (2.58)	0.878 *** (3.38)	0.058 * (1.93)	0.072 ** (2.40)
ROAPER	-6.808 *** (-4.89)	-5.969 *** (-4.24)	-0.926 *** (-3.84)	-0.817 *** (-3.35)	-7.592 *** (-5.16)	-6.702 *** (-4.45)	-0.497 *** (-2.87)	-0.441 ** (2.53)
BOMARK	0.247 *** (3.70)	0.286 *** (4.26)	0.023 ** (1.96)	0.029 ** (2.52)	0.263 *** (3.75)	0.326 *** (4.49)	0.016 * (1.91)	0.199 ** (2.39)
INDERPER	-0.197 (-0.13)	0.153 (0.10)	-0.063 (-0.23)	-0.010 (-0.04)	-0.943 -0.57)	-0.408 (-0.24)	-0.119 (-0.61)	-0.093 (-0.47)
EQUINAT	0.215 (0.78)	0.338 (1.21)	-0.042 (-0.88)	-0.025 (-0.52)	-0.205 (-0.70)	-0.083 (-0.28)	-0.030 (-0.86)	-0.020 (-0.59)
CONSTANT	-19.221 *** (-6.58)	3.930 *** (4.43)	-3.578 *** (-7.08)	0.117 (0.76)	-20.800 *** (-6.75)	10.114 *** (10.65)	-1.774 *** (-4.88)	0.121 (1.10)
<b>Fixed Effects</b>	Ŷ	Y	Y	Y	Y	Y	Y	Ŷ
Observations	2559	2558	2559	2558	2559	2558	2559	2558
R <sup>2</sup> -adjusted	0.119	0.088	0.109	0.086	0.141	0.091	0.127	0.094
F-test	4.78 ***	5.02 ***	3.19 ***	3.34 ***	4.20 ***	4.30 ***	2.77 ***	2.88 ***

Table 5. Executive incentives and the inter-temporal tradeoffs for CSR preferences under earnings pressure.

**Notes:** All models include year and industry fixed effects. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

Furthermore, the coefficients also show that executives with short-term incentives tend to take institutional CSR items as "insurance" for negative evens or earnings obsession. In particular, executives

with short-term incentives may reduce CSR concerns activities under earnings pressure; thus, they could gain more "community reputation" or moral capital at less expense. The results provide the evidence that goodwill-related activities may attenuate the reaction of stakeholders to poor performance and adverse events, consistent with Lins and Servaes [59] and Flammer [60]. For example, the interaction terms between executive short-incentives (SHOEXE) and earnings pressure (EARPRE) are significantly negative with technical CSR strengths items. The coefficient of interaction terms is -0.158 with a t-statistics of -1.83 and 10% statistical significance in column (6), and the coefficient is -0.136 with a t-statistics of -1.69 and 10% statistical significance in column (2).

What is more, it is worth noting that executives with long-term incentives would like to make inter-temporal behaviors of weakening CSR strengths items under earnings pressure. Compared with other long-term strategies, reducing institutional CSR strengths could significantly promote the increase of short-term performance rather than introducing a negative implication. Although executives' long-term incentives focus on the sustainable development of enterprises by practicing long-term strategies, they still pay attention to the fluctuations of the stock price. This finding is consistent with Antia et al. [25].

#### 4.3.2. Hypocrisy Evidence on the Inter-Temporal Tradeoffs for CSR types

Table 6 presents the results for hypocrisy and the inter-temporal tradeoffs for different CSR activities under earnings pressure. Columns (1) to (4) of Table 6 show the results of the inter-temporal tradeoffs of executives with short-term incentives in the context of low earnings management ("EM low"). The interaction terms between executive short-incentives (SHOEXE) and earnings pressure (EARPRE) show positive relationships with the net CSR scores, including NETINCSR and NETECSR, while statistical significance is more than 10%. Besides, columns (5) to (6) of Table 6 show the results in the context of high earnings management ("EM high"). The interaction terms between executive short-incentives (SHOEXE) and earnings management ("EM high"). The interaction terms between executive short-incentives (SHOEXE) and earnings pressure (EARPRE) show positive relationships with the net CSR scores, including NETINCSR and NETECSR. The coefficient of the interaction term is -3.731 with a t-statistics of -1.65 and 10% statistical significance in column (6), and the coefficient is -4.741 with a t-statistics of -1.99 and 10% statistical significance in column (8).

		Dependent Variable						
		Subsamp	ole: EM Low	r		Subsamp	le: EM Higł	ı
	NEINCSR		NETECSR		NEI	NCSR	NETECSR	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
CHOEVE	0.973 **	0.831	2.633 ***	2.455 ***	1.072 **	1.061 **	1.423 ***	1.408 ***
SHOEXE	(1.98)	(1.63)	(5.15)	(4.63)	(2.47)	(2.45)	(3.11)	(3.10)
SHOEXE		-2.751		-3.439		-3.731 *		-4.741 **
<b>×EARPRE</b>		(-1.05)		(-1.26)		(-1.65)		(-1.99)
EADDDE	0.559	1.646	-0.575	0.782	1.055	2.508 *	-1.111	0.736
EAKFKE	(0.58)	(1.16)	(-0.57)	(0.53)	(1.13)	(1.96)	(-1.13)	(0.55)
Control vars	Y	Y	Y	Y	Y	Y	Y	Y
<b>Fixed Effects</b>	Y	Y	Y	Y	Y	Y	Y	Y
Observisions	790	790	790	790	731	731	731	731
R <sup>2</sup> -adjusted	0.010	0.032	0.094	0.092	0.065	0.059	0.043	0.040
F-test	2.33 ***	2.33 ***	2.25 ***	2.26 ***	2.09 ***	2.10 ***	2.12 ***	2.15 ***

Table 6. Hypocrisy and the inter-temporal tradeoffs for CSR types under earnings pressure.

**Notes:** All models include year and industry fixed effects. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

What's more, the coefficients of the interaction terms between executives with short-term incentives (SHOEXE) and earnings pressure (EARPRE) on technical CSR items (NETECSR) are higher than the coefficients of the interaction term on institutional CSR items (NEINCSR). The coefficients indicate

that executives with short-term incentives are inclined to take technical CSR-related activities as "hypocritical tools" compared with institutional CSR -related activities. The results supplement the empirical evidence that executives take CSR-related actions as a tool to disguise hypocrisy with self-interest embedded in benevolence.

# 4.4. Do Institutional Investors Supervise the Inter-Temporal Tradeoffs for CSR Types of Executives under *Earnings Pressure*?

#### 4.4.1. Main Results on Institutional Investors and CSR Types of Executives

In this section, we would like to answer three questions. The first is whether institutional investors monitor executives to different CSR activities in the capital market. The second is how institutional investors with disparate characters affect the executives' choices on various CSR activities. The third is whether there is a significant change for executives' decisions on different CSR activities when considering the monitoring of institutional investors.

Table 7 presents the results as follows. The coefficients reveal that the transient institutional investors (TRANINS) with stocks holding in numerous enterprises and frequently trading in and out of enterprises show opposition to CSR-related activities. The interaction terms between short-incentives (SHOEXE) and transient institutional investors (TRANINS) are significantly negative with institutional CSR items (NETINCSR) and technical CSR items (NETECSR). This result could be supported by the research of Fich et al. [36], who point out that transient investors frequently trade in and out of firms on the base of change in stock market value, and they are less likely to perform corporate governance. However, the absolute values of coefficients in the context of the S-TIH subsample are higher than the L-TIH subsample. For example, the coefficient of the interaction term is -0.128 with a t-statistics of -4.14 and 1% statistical significance in column (1), which is more than the coefficient in column (5). Similarly, the coefficient of the interaction term is -0.131 with a t-statistics of -3.97 and 1% statistical significance in column (2), which is higher than the coefficient in column (6).

In addition, the dedicated institutional investors (DEDIINS) with stocks holding in a small number of enterprises and non-frequently trading have incentives to supervise hypocritical behaviors, which is consistent with Cho et al. [40]. The coefficients of interaction terms between executives with short-term incentives (SHOEXE) and dedicated institutional investors (DEDIINS) is not significant, compared with the coefficient in column (4). For example, the coefficient of interaction terms is -0.205 with a t-statistics of -0.73 and more than 10% significance in column (3), and the coefficient is 0.468 with a t-statistics of 1.71 and 10% statistical significance in column (4). Although the coefficients of interaction terms are not significant, the dedicated institutional investors (DEDIINS) effectively balance the opposition to executives with short-term incentives' choice on CSR-related activities. Especially, the coefficient in column (4) shows that the dedicated institutional investors (DEDIINS) could promote the technical CSR items of enterprises.

#### 4.4.2. Institutional Investors and the Inter-Temporal Tradeoffs for CSR Types

Table 8 presents the results for institutional investors and the inter-temporal tradeoffs of executives under earnings pressure in the context of investment horizon subsamples. The results provide additional evidence that institutional investors with the long-term horizon (LONINS) can effectively constrain executive hypocrisy. In particular, the investment horizon of institutional investors shows an inverse relationship with the level of the executives' support for CSR-related activities. To dig deeper into this effect, this section constructs the interaction term, including the executive incentives and earnings pressure in the context of investment horizon subsamples. This method helps to investigate the change for CSR preferences of executives when considering the monitoring of institutional investors.

				Depender	nt Variable			
		Subsamp	le: S-TIH			Subsamp	le: L-TIH	
	NEINCSR	NETECSR	NEINCSR	NETECSR	NEINCSR	NETECSR	NEINCSR	NETECSR
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
SHOEXE	1.346 *** (9.1)	1.793 *** (11.40)	-0.137	-1.818 ** (-2.05)	1.682 *** (11.55)	2.018 ***	1.077 (1.05)	0.996 (0.76)
SHOEXE ×TRANINS	-0.128 ***	-0.131 ***	( 0.10)	( 2.00)	-0.073 **	-0.043 **	(100)	(011-0)
SHOEXE ×TRANINS	-0.128 *** (-4.14)	-0.131 *** (-3.97)			-0.073 ** (-4.46)	-0.043 ** (-2.43)		
SHOEXE ×DEDIINS	1 / 11 444	1 50 444	0.205 (0.73)	0.468 * (1.71)	0.040 ***	0.466.5	0.049 (0.31)	0.062 (0.44)
TRANINS	(3.69)	(3.33)	2 070		(4.00)	(1.84)	0.445	0.000
DEDIINS			-2.879 (-0.70)	-6.807 * (-1.70)			-0.665 (-0.29)	-0.900 (-0.43)
CASHRA	0.157 (1.25)	0.188 (1.40)	5.245 (1.30)	5.493 (1.40)	0.141 (1.12)	0.171 (1.27)	0.441 (0.12)	-0.098 (-0.03)
DUTYMER	0.078 (0.62)	-0.117 (-0.87)	0.117 (0.12)	1.823 * (1.86)	0.102 (0.84)	-0.093 (-0.71)	-0.883 (-1.13)	0.854 (1.22)
SEPARATE	0.005 (0.43)	-0.004 (-0.32)	0.154 (1.07)	-0.150 (-1.07)	0.004 (0.34)	-0.006 (-0.48)	-0.137 * (-1.88)	-0.033 (-0.51)
BALANCE	0.778 *** (5.17)	0.729 *** (4.55)	-0.419 (-0.49)	-0.437 (-0.52)	0.682 *** (4.36)	0.729 *** (4.60)	0.924 (1.29)	0.246 (0.38)
ROAPER	-2.106 ** (-2.23)	-3.310 *** (-3.29)	-7.931 (-0.89)	0.268 (0.03)	-2.463 *** (-2.63)	-4.458 *** (-4.42)	-6.448 (-0.98)	4.632 (0.79)
BOMARK	0.370 *** (3.91)	0.362 *** (3.59)	0.075 (0.10)	1.039 (1.37)	0.331 *** (3.48)	0.309 *** (3.02)	-0.011 (-0.02)	-0.378 (-0.65)
INDERPER	-1.369 (-1.29)	-1.175 (-1.04)	-1.731 (-0.28)	-9.684 (-1.61)	-1.340 (-1.27)	-0.914 (-0.81)	-12.190 ** (-2.17)	-8.228 * (-1.64)
EQUINAT	0.105 (0.54)	-0.062 (-0.30)	-0.319 (-0.32)	-0.387 (-0.40)	-0.001 (-0.00)	-0.149 -0.73)	-2.239 (-0.89)	0.032 (-0.01)
CONSTANT	-13.722 *** (-6.31)	-14.685 *** (-6.34)	-10.909 (-0.57)	15.209 (0.81)	-18.214 ** (2.145)	-17.719 ** (-7.68)	-9.125 (-0.43)	-5.360 (-0.28)
<b>Fixed Effects</b>	Y	Y	Y	Y	Y	Y	Y	Y
Observations	4366	4366	294	294	4491	4491	275	275
R <sup>2</sup> -adjusted	0.137	0.140	0.037	0.036	0.123	0.117	0.077	0.049
F-test	4.88 ***	5.14 ***	2.20 ***	2.70 ***	5.06 ***	5.30 ***	3.44 ***	4.22 ***

**Table 7.** Institutional investors and CSR types.

**Notes:** All models include year and industry fixed effects. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

Table 8. Institutional investors and the inter-temporal tradeoffs for CSR preferences of executives.

		Dependent Variable						
		Subsamp	le: S-TIH			Subsamp	le: L-TIH	
	NEINCSR		NETECSR		NEIN	NCSR	NETECSR	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
SHOEXE	1.571 *** (5.32)	1.567 *** (5.32)	0.663 *** (8.43)	2.659 *** (8.43)	0.601 ** (2.03)	0.598 ** (2.01)	1.319 *** (4.21)	1.287 *** (4.11)
SHOEXE ×EARPRE	, , ,	-0.304 *	, ,	-0.298 * (-1.74)	, , ,	-0.015 (-0.11)		-0.234 (-1.59)
EARPRE	-0.556 (-0.86)	0.197 (0.26)	-0.756 (-1.10)	-0.019 (-0.02)	0.786 (1.37)	0.820 (1.25)	0.185 (0.31)	0.715 (1.04)
<b>Control Vars</b>	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ
<b>Fixed Effects</b>	Y	Y	Y	Y	Y	Y	Y	Y
Observations	1223	1223	1223	1223	1305	1305	1305	1305
R <sup>2</sup> -adjusted	0.077	0.079	0.130	0.134	0.041	0.041	0.023	0.025
F-test	2.83 ***	2.84 ***	2.47 ***	2.47 ***	3.89 ***	3.89 ***	3.90 ***	3.91 ***

**Notes:** All models include year and industry fixed effects. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

Columns (1) to (4) of Table 8 show that the interaction terms between executives with short-term incentives (SHOEXE) and earnings pressure (EARPRE) are significantly negative with the net institutional CSR items (NEINCSR) and the net technical CSR items (NETECSR), respectively. For example, the coefficient of the interaction on the net institutional CSR scores (NEINCSR) is -0.304 with a t-statistics of -1.90 and 10% statistical significance in column (2). The coefficient of the interaction on the net technical CSR scores (NETECSR) is -0.298 with a t-statistics of -1.74 and 10% statistical significance in column (4). The coefficients indicate that executives with short-term incentives practice inter-temporal tradeoffs under earnings pressure if institutional investors have a short investment horizon.

Columns (5) to (8) of Table 8 show that the interaction terms between executives with short-term incentives (SHOEXE) and earnings pressure (EARPRE) show an adverse relationship with the net institutional CSR items (NEINCSR) and the net technical CSR items (NETECSR), respectively. Although the coefficient is not significant at the statistical level, the institutional investors with long-term horizon (LONINS) effectively balance the inter-temporal tradeoffs of executives. The result also provides additional evidence that the dedicated institutional investors (DEDIINS) with long-horizon could play important monitoring roles in correcting the CSR strategy of executives.

#### 4.5. Robustness Tests

#### 4.5.1. Endogeneity

The endogeneity concern could be the main problem in the above-related research. To rule out reverse causality, we employ the instrumental variables method of system moment estimation to avoid the endogeneity concern, and the results are presented in Appendix D. The most challenging task is the selection of instrumental variables, which require the strict method definition. In this paper, we employ the lagged stock concentration and the lagged investment concentration of institutional investors as the instrumental variables. The instrumental variables are likely to affect the size and duration of institutional investors' holdings, but impossibly directly affect the institutional and technical CSR scores. In the section of Table A2 in Appendix D, the study regresses the net CSR scores, including NETINCSR and NETTECSR, on institutional investors according to their portfolio durations and average holding time based on the model of two-step Generalized Method of Moments (GMM). The coefficient reveals that both transient and dedicated institutional investors tend to choose behavioral models that not supporting CSR activities. Especially, the strength of the opposition coefficients is significantly negative with the average holding time based on the perspective of the investment horizon. Above all, the results provide additional evidence that institutional investors have a causal effect on CSR, and the results in this study are robustness.

#### 4.5.2. Alternative Measure for Executive Incentives

Although the selection and measurement of variables used in this study are discussed in detail, this study rechecks the robustness of our results employing the alternative measure for executive incentives. This research employs the absolute values of the proxies to describe executive incentives. Namely, when the executives are honored with compensation incentives or equity incentives by shareholders, the occurring of principal-agent problems would be decreased. Furthermore, in order to make sure our conclusion robustness, the executive incentives of the relative index are constructed. To classify executives according to the relative index, this study sorts the executive incentives into the relative long-term incentives and the relative short-term incentives by taking the median as the grouping criterion.

#### 4.5.3. Further Robustness Tests

To ensure the robustness of the conclusions, this paper practices three basic checks. First, each regression with different samples by splitting the time, such as the period from 2010 to 2017,

and the period from 2013 to 2017 is running. Second, the industry fixed effects based on different industry classification codes are recreated, including the companies less than two years. Third, the models, including firm fixed effects and province fixed effects, are re-estimated. The conclusions of this paper are robust to all three tests.

#### 5. Conclusions and Implications

#### 5.1. Conclusions

This study is a conductive supplement for the strand of literature on CSR activities in China as an emerging market. We conceptualize the impact of the executive incentives on CSR activities, then extend the framework to the case that firms are under pressure from the financial analyst and institutional investors. A series of robustness tests provide robust evidence for our conclusions. This paper has the following findings.

First, executives with different incentive preferences choose different CSR-related activities to enhance the sustainable development of the enterprises, rather than hinder firm growth due to their self-interests and myopic behaviors. Especially, executives with long-term incentives prefer to choose institutional CSR items to enhance the harmonious relationship with the community, while those with short-term incentives tend to choose technical CSR items. Moreover, executives with short-term incentives may disguise hypocrisy through practicing CSR-related activities and embed self-interest into charity behaviors. Furthermore, although executives with long-term incentives also perform earnings management for current stock price stability, they would like to practice CSR-related activities, especially institutional CSR.

Second, executives could perform inter-temporal tradeoffs for short-term financial performance in the earnings pressure context. This article suggests that executives with short-term incentives are more likely to practice inter-temporal tradeoffs under earnings pressure. The inter-temporal tradeoffs mainly appear in technical CSR items, not obviously for institutional CSR items. Furthermore, although executives with long-term incentives practice the inter-temporal tradeoffs, the intensity of their motivation is significantly weaker than that of executives with short-term incentives. It shows that executives with long-term incentives could pay more attention to the long-term value growth of enterprises while maintaining the short-term stock price stability.

Third, institutional investors play an important role in supervising the hypocrisy of executives and help to improve the inner governance of firms. Moreover, institutional investors with disparate features show different governance effects. Dedicated institutional investors help to restrain the hypocrisy behaviors of executives, who hold stocks in a small number of companies and non-frequently trade. Executives are less likely to practice inter-temporal tradeoffs in the earnings pressure context if institutional investors have a long-term investment horizon. In conclusion, dedicated institutional investors with long-term horizon show better governance effect in supervising and correcting behaviors of executives.

#### 5.2. Managerial Implications

Our findings provide managerial implications for executives and investors. This article suggests that, in China, listed company investors should focus on the incentive preferences of executives to responsible investments, because the CSR-related activities may be a kind of corporate hypocrisy with self-interest embedded in benevolence. For example, executives with short-term incentives are inclined to take technical CSR-related activities as "hypocritical tools" rather than institutional CSR-related activities. These hypocritical behaviors could be harmful to the sustainable development of firms and be utilized to achieve executives' self-interests at the expense of shareholders' interests. Moreover, investors should pay attention to the earnings pressure on executives, because executives may practice inter-temporal tradeoffs to promote short-term performance by cutting long-term investment. These inter-temporal tradeoffs behaviors are not conducive to the sustainable development of firms due to their strategic uncertainty. Also, institutional investors play an important role in supervising

and regulating the hypocrisy behaviors of executives. As a recent emerging supervision power in China, the types of investors may significantly affect the executive strategic decision and financial performance of the firm due to their strengths in capital and information. Therefore, it is helpful to improve the ownership structure of listed firms to enhance the governance effects of institutional investors, such as increasing shares of dedicated investors with long-term horizon.

Additionally, the results of this study can stimulate the interest of policymakers since it provides some evidence about the paradigms of executives and the incentive preferences of executives to CSR in the pressure situations. Our findings provide policy implications for policymakers in China as follows. First, some corporate governance regulations should be issued to perfect the institutional design of listed firms and to regulate the behaviors of executives with disparate incentives. Second, it's urgent to reduce information asymmetry between the firms and the investors to further promote transparency both in financial and non-financial information of listed firms in the emerging market. Third, because of the lack of reasonable policy guidance for institutional investors in emerging markets, it is beneficial to design reasonable multi-level guidance for different institutional investors.

#### 5.3. Limitations and Future Research

This study presents the tests of the impact of the executives with different incentive preferences on CSR activities, then extend the framework in the pressure situations. Although the results of this article are valuable and provide some meaningful implications, several aspects limit the current research. First, the executive incentives as the proxy are utilized to measure the behavior paradigms of top-level executives, and it can't adequately represent the motivation of executives' behaviors due to the unreported compensations and hidden benefits existing. Therefore, the future search should construct a more effective proxy to executive incentives, which consider the unreported compensations and covered benefits. Second, the earning pressure is measured by the gap between the consensus of financial analysts and the potential performance of firms evaluated at the beginning of the reporter term. This measure ignores the changes in the market environment over the accounting year, which may affect the measure of this proxy. Future research may introduce the dynamic environment into the analysis framework. Finally, more dimensions for CSR strategies should be considered, which is constructive to enrich the results and implications of research in the emerging markets.

**Author Contributions:** The authors contributed equally to this work. All the authors contributed to the conceptualization, formal analysis, investigation, methodology, writing of the original draft, and review and editing. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the National Social Science Fund of China (19VHQ002).

Conflicts of Interest: The authors declare no conflict of interest.

#### Appendix A

Abbreviation	Variable	Description
		Corporate responsibility variables
INCSRS	ICSR Strenths	The sum score of the positive items under the dimensions of community relations, environmental stewardship, and diversity. Source: CCSR database.
INCSRC	ICSR Concerns	The sum score of the negative items under the dimensions of community relations, environmental stewardship, and diversity. Source: CCSR database.
NETINCSR	ICSR Net	The score of INCSRP minus the score of INCSRN
TECSRS	TCSR Stengths	The sum score of the positive items under the dimensions of employee relations, product safety, and governance. Source: CCSR database.
TECSRC	TCSR Concerns	The sum score of the negative items under the dimensions of employee relations, product safety, and governance. Source: CCSR database.
NETTECSR_	TCSR Net	The score of TECSRP minus the score of TECSRN

Table A1. Definition of variables.

Table A1. Cont.					
Abbreviation	Variable	Description			
		Independent variable			
LONEXE	Long Executive Incentives	The natural logarithm of the total compensation of the top three senior executives plus one. Source: WIND database.			
SHOEXE	Short Executive Incentives	The average market value of restricted stock and unexercised stock options, owned by the senior executives. Source: WIND database.			
DEDIINS	Dedicated Institution	The shareholding value in the target firm is in the top 10% of their portfolio. Source: WIND database.			
TRANINS	Transient Institution	The shareholding value in the target firm is not in the top 10% of their portfolio. Source: WIND database.			
LONDEDI	Long Dedicated Institution	Long monitoring institution indicates the dedicated institutional investors with long-term horizon. Source: WIND database.			
SHODEDI	Short Dedicated Institution	Short monitoring institution indicates the dedicated institutional investors with short-term horizon. Source: WIND database.			
LONTRAN	Long Transient Institution	Long monitoring institution indicates the transient institutional investors with long-term horizon. Source: WIND database.			
SHOTRAN	Short Transient Institution	Short monitoring institution indicates the transient institutional investors with short-term horizon. Source: WIND database.			
		Moderator variable			
EARPRE	Earnings Pressure	The difference between the consensus of analysts' earnings forecast and firms' potential earnings evaluated at the beginning of the reporter term. Source: CSMAR database.			
LONINS	Long-incentives Institution	Dummy variables that take the value of 1 if the average holding time is greater than the average position of institutional investors in each industry each year. Source: WIND database.			
SHOINS	Short-incentives Institution	Dummy variables that take the value of 1 if the average holding time is less than the average position of institutional investors in each industry each year. Source: WIND database.			

SHOINS	Short-incentives Institution	less than the average position of institutional investors in each industry each year. Source: WIND database.						
	Control variable							
CASHRA	Cash Holding Ratio	The ratio of cash holding over total book assets. Source: CSMAR database.						
DUTYMER	Duty Merging	Dummy variable that takes the value of 1 if chairman and CEO are the same people; otherwise, the value of 0. Source: CSMAR database.						
SEPERATE	Separation of Powers	The difference between the ultimate controller's control and cash flow rights. Source: CSMAR database.						
BALANCE	Equity Balance	The ratio of the shareholding ratio of the second to the tenth largest shareholder to the largest shareholder. Source: CSMAR database.						
ROAPER	Roa Performance	The ratio of net income over total book assets. Source: CSMAR database.						
BOMARK	Book Market Ratio	The ratio of equity book value over equity market value. Source: CSMAR database.						
INDEPER	Independent Directors Ratio	The ratio of the number of independent directors over the total number of board members. Source: CSMAR database.						
EQUNAT	Equity Nature	Dummy variable that takes the value of 1 if the enterprise is state-owned enterprises, otherwise, the value of 0. Source: CSMAR database.						

### Appendix **B**

Following Gaspar et al. [48], this section constructs the proxy of turnover rate (Turnover) in three steps. First, calculating *Turnaround* indicating the change of the investors' portfolios of investor *j* and quarter *t* for two quarters as

$$Turnaround_{i,t} = \frac{\sum_{i \in Q} \left| N_{i,j,t} P_{i,t} - N_{i,j,t-1} P_{i,t-1} - N_{i,j,t-1} (P_{i,t} - P_{i,t-1}) \right|}{\sum_{i \in Q} \frac{1}{2} (N_{i,j,t} P_{i,t} + N_{i,j,t-1} P_{i,-1})}$$
(A1)

where Q denotes the set of stocks held by investor j,  $P_{i,t}$  is the price of stock i at quarter t, and  $N_{i,j,t}$  is the number of stock *i* held by investor *j* at quarter *t* in the reporting term. Second, averaging the four Turnaround annually to calculate the portfolio *Turnaround* of investors. Third, constructing firm-level *Turnaround* as the weighted average of the investors' portfolios *Turnover*.

#### Appendix C

According to the method of the Jones model [55,56], this section calculates discretionary accruals in two steps. First, running the following cross-sectional model by each year and CSRC industry classification to estimate the parameters  $\alpha_1$ ,  $\alpha_2$ , and  $\alpha_3$ 

$$\frac{TA_{i,t}}{A_{i,t-1}} = \alpha_1 \frac{1}{A_{i,t-1}} + \alpha_2 \frac{\Delta REV_{i,t}}{A_{i,t-1}} + \alpha_3 \frac{PPE_{i,t}}{A_{i,t-1}} + \varepsilon_{i,t}$$
(A2)

where  $TA_{i,t}$  equals net income minus cash flow from operations of enterprise *i* at time *t*,  $\Delta REV_{i,t}$  is the changes in sales revenue,  $PPE_{i,t}$  is gross property, plant, and equipment, and  $A_{i,t-1}$  are total book assets of time *t*-1. Second, using the estimated parameters  $\alpha_1$ ,  $\alpha_2$ , and  $\alpha_3$  to calculated discretionary accruals of firm *i* at time *t* as

$$DA_{i,t} = \frac{TA_{i,t}}{A_{i,t-1}} - \alpha_1 \frac{1}{A_{i,t-1}} - \alpha_2 (\frac{\Delta REV_{i,t}}{A_{i,t-1}} - \frac{\Delta AR_{i,t}}{A_{i,t-1}}) - \alpha_3 \frac{PPE_{i,t}}{A_{i,t-1}}$$
(A3)

where  $\Delta AR_{i,t}$  is the change in receivables of firm *i* at time *t*. As all variables are scaled, the enterprise's discretionary accruals are indicated as a percentage of the enterprise's total book assets.

#### Appendix D

	Dependent Variable							
	NEINCSR	NETECSR	NEINCSR	NETECSR	NEINCSR	NETECSR	NEINCSR	NETECSF
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
SHOTRAN	-1.222 ***	-1.609 ***						
	(-10.29)	(-11.98)						
LONTRAN			-0.625 ***	-0.794 ***				
			(-10.02)	(-11.22)				
SHODEDI					-2.561 **	-3.100 ***		
					(-2.53)	(-2.63)		
LONDEDI							-1.168 **	-1.366 **
	0.054.333		0.101.444	0.4.00 ***	0.005 **	0.0004	(-2.19)	(-2.48)
CASHRA	0.074 ***	0.057 **	0.134 ***	0.133 ***	-0.225 **	-0.2224 **	-0.156 *	-0.170 *
	(3.08)	(2.09)	(5.34)	(4.66)	(-2.30)	(-1.96)	(-1.70)	(0.094)
DUTYMER	0.274 *	0.103	0.171	0.002	1.955	2.977 **	0.936	0.909
	(1.67)	(0.55)	(1.00)	(0.01)	(1.63)	(2.13)	(1.11)	(1.04)
SEPARATE	0.023 **	0.022 **	0.023 ***	0.022 **	-0.025	-0.072	0.002	-0.032
	(2.42)	(2.10)	(2.33)	(1.96)	(-0.46)	(-1.12)	(0.05)	(-0.64)
BALANCE	1.133 ***	1.419 ***	1.289 ***	1.639 ***	1.132 *	1.192	1.778 **	2.124 ***
	(8.68)	(9.60)	(9.03)	(10.13)	(1.71)	(1.55)	(2.41)	(2.78)
ROAPER	13.049 ***	16.440 ***	17.565 ***	21.414 ***	9.098	10.444	9.529	10.523
	(8.75)	(9.74)	(9.48)	(10.20)	(1.21)	(1.20)	(1.25)	(1.33)
BOMARK	0.069	0.078	0.023	0.029	0.657	0.183	1.541 ***	1.333 *
	(0.75)	(0.77)	(0.25)	(0.28)	(1.06)	(0.25)	(2.97)	(2.48)
INDERPER	3.191 ***	3.160 ***	4.447 ***	4.789 ***	-3.071	-0.483	0.647	1.996
	(3.00)	(2.63)	(3.99)	(3.79)	(-0.50)	(-0.07)	(0.11)	(0.34)
EQUINAT	-0.376 ***	-0.356 ***	-0.315 ***	-0.282 ***	1.065 *	1.848 **	0.755	1.182 **
	(-3.38)	(-3.83)	(-2.70)	(-2.13)	(1.65)	(2.45)	(1.43)	(2.16)
CONSTANT	5.302 ***	10.621 ***	5.529 ***	10.779 ***	5.753 *	8.146 **	4.605	9.206 **
	(8.93)	(15.82)	(8.80)	(15.14)	(1.67)	(2.04)	(1.42)	(2.74)
LM statistics	291.28 ***	291.28 ***	253.65 ***	253.65 ***	7.25 **	7.248 **	9.623 ***	9.623 ***
Sargan	0.047	1.639	0.205	0.349	0.493	0.003	0.074	0.006
statistics	(0.828)	(0.200)	(0.651)	(0.555)	(0.483)	(0.957)	(0.786)	(0.936)
Notes: All models include year and industry fixed effects. *, **, and *** denote statistical significance at the 10%, 5%,								

Table A2. Instrumental variable approach with two-step GMM estimation.

**Notes:** All models include year and industry fixed effects. \*, \*\*, and \*\*\* denote statistical significance at the 10%, 5%, and 1% levels, respectively.

#### References

- 1. See, G. Harmonious society and Chinese CSR: Is there really a link? J. Bus. Ethics 2009, 89, 1–22. [CrossRef]
- 2. Kim, S.; Ji, Y. Chinese consumers' expectations of corporate communication on CSR and sustainability. *Corp. Soc. Responsib. Environ. Manag.* **2017**, *24*, 570–588. [CrossRef]
- 3. Anner, M.S. Social activism and practice diffusion: How activist tactics affect non-targeted organizations. *Adm. Sci. Q.* **2015**, *60*, 300–332.
- 4. McDonnell, M.H.; King, B.G.; Soule, S.A. A dynamic process model of private politics: Activist targeting and corporate receptivity to social challenges. *Am. Sociol. Rev.* **2015**, *80*, 654–678. [CrossRef]
- Freeman, R.E.; Wicks, A.C.; Parmar, B. Stakeholder theory and "the corporate objective revisited". *Organ. Sci.* 2004, 15, 364–369. [CrossRef]
- 6. Porter, M.E.; Kramer, M.R. The link between competitive advantage and corporate social responsibility. *Harv. Bus. Rev.* **2006**, *84*, 78–92.
- Jo, H.; Harjoto, M.A. The causal effect of corporate governance on corporate social responsibility. *J. Bus. Ethics* 2012, 106, 53–72. [CrossRef]
- 8. Adhikari, B.K. Causal effect of analyst following on corporate social responsibility. *J. Corp. Financ.* **2016**, *41*, 201–216. [CrossRef]
- 9. Malik, F.; Wang, F.; Naseem, M.A.; Ikram, A.; Ali, S. Determinants of Corporate Social Responsibility Related to CEO Attributes: An Empirical Study. *Sage Open* **2020**, *10*, 2158244019899093. [CrossRef]
- 10. McClelland, P.L.; Barker, V.L., III; Oh, W.Y. CEO career horizon and tenure: Future performance implications under different contingencies. *J. Bus. Res.* **2012**, *65*, 1387–1393. [CrossRef]
- 11. Abernethy, M.A.; Jiang, L.; Kuang, Y.F. Can organizational identification mitigate the CEO horizon problem? *Account. Organ. Soc.* **2019**, *78*, 101056. [CrossRef]
- 12. Womack, K.L. Do brokerage analysts' recommendations have investment value? *J. Financ.* **1996**, *51*, 137–167. [CrossRef]
- 13. Fogarty, T.J.; Rogers, R.K. Financial analysts' reports: An extended institutional theory evaluation. *Account. Organ. Soc.* **2005**, *30*, 331–356. [CrossRef]
- 14. Groysberg, B.; Lee, L.E.; Nanda, A. Can they take it with them? The portability of star knowledge workers' performance. *Manag. Sci.* 2008, 54, 1213–1230.
- 15. Zhang, Y.; Gimeno, J. Earnings pressure and long-term corporate governance: Can long-term-oriented investors and executives reduce the quarterly earnings obsession? *Organ. Sci.* **2016**, *27*, 354–372. [CrossRef]
- 16. Connelly, B.L.; Tihanyi, L.; Certo, S.T.; Hitt, M.A. Marching to the beat of different drummers: The influence of institutional owners on competitive actions. *Acad. Manag. J.* **2010**, *53*, 723–742. [CrossRef]
- 17. Qian, C.; Lu, L.Y.; Yu, Y. Financial analyst coverage and corporate social performance: Evidence from natural experiments. *Strateg. Manag. J.* **2019**. [CrossRef]
- 18. Finkelstein, S.; Hambrick, D.C. Chief executive compensation: A synthesis and reconciliation. *Strateg. Manag. J.* **1988**, *9*, 543–558. [CrossRef]
- 19. Silberzahn, C.; Arregle, J.L. The career-horizon problem in capital investments for lone-founder and long-tenure acquirer CEOs in their final career stage. *Strateg. Organ.* **2019**, *17*, 334–362. [CrossRef]
- Marinovic, I.; Varas, F. CEO Horizon, Optimal Pay Duration, and the Escalation of Short-Termism. *J. Financ.* 2019, 74, 2011–2053. [CrossRef]
- 21. Fabrizi, M.; Mallin, C.; Michelon, G. The role of CEO's personal incentives in driving corporate social responsibility. *J. Bus. Ethics* **2014**, *124*, 311–326. [CrossRef]
- 22. Hong, B.; Li, Z.; Minor, D. Corporate governance and executive compensation for corporate social responsibility. *J. Bus. Ethics* **2016**, *136*, 199–213. [CrossRef]
- 23. Souder, D.; Bromiley, P. Explaining temporal orientation: Evidence from the durability of firms' capital investments. *Strateg. Manag. J.* 2012, *33*, 550–569. [CrossRef]
- 24. Oh, W.Y.; Chang, Y.K.; Cheng, Z. When CEO career horizon problems matter for corporate social responsibility: The moderating roles of industry-level discretion and blockholder ownership. *J. Bus. Ethics* **2016**, *133*, 279–291. [CrossRef]
- 25. Antia, M.; Pantzalis, C.; Park, J.C. CEO decision horizon and firm performance: An empirical investigation. *J. Corp. Financ.* **2010**, *16*, 288–301. [CrossRef]

- 26. Zhang, Y.; Gimeno, J. Earnings pressure and competitive behavior: Evidence from the US electricity industry. *Acad. Manag. J.* **2010**, *53*, 743–768. [CrossRef]
- Hoskisson, R.E.; Hitt, M.A.; Johnson, R.A.; Grossman, W. Conflicting voices: The effects of institutional ownership heterogeneity and internal governance on corporate innovation strategies. *Acad. Manag. J.* 2002, 45, 697–716.
- 28. Zhang, Y.; Gong, Y. Stock Return or Sales Growth? Multiple Performance Feedback and Strategic Investments Under Securities Analysts' Earnings Pressure. *J. Manag. Stud.* **2018**, *55*, 1356–1385. [CrossRef]
- Schulz, A.C.; Wiersema, M.F. The impact of earnings expectations on corporate downsizing. *Strateg. Manag. J.* 2018, 39, 2691–2702. [CrossRef]
- 30. Duong, C.; Pescetto, G. Overvaluation and earnings management: Does the degree of overvaluation matter? *Account. Bus. Res.* **2019**, *49*, 121–146. [CrossRef]
- 31. Lee, C.Y.; Chang, H.Y. How do the combined effects of CEO decisionhorizon and compensation impact the relationship between earnings pressure and R&D retrenchment? *Technol. Anal. Strateg. Manag.* **2014**, *26*, 1057–1071.
- 32. Lian, C.; Ma, Y.; Wang, C. Low Interest Rates and Risk-Taking: Evidence from Individual Investment Decisions. *Rev. Financ. Stud.* 2018, *32*, 2107–2148. [CrossRef]
- Gillan, S.L.; Starks, L.T. The evolution of shareholder activism in the United States. J. Appl. Corp. Financ. 2007, 19, 55–73. [CrossRef]
- 34. Alvarez, R.; Jara, M.; Pombo, C. Do institutional blockholders influence corporate investment? Evidence from emerging markets. *J. Corp. Financ.* **2018**, *53*, 38–64. [CrossRef]
- 35. Motta, E.M.; Uchida, K. Institutional investors, corporate social responsibility, and stock price performance. *J. Jpn. Int. Econ.* **2018**, *47*, 91–102. [CrossRef]
- 36. Fich, E.M.; Harford, J.; Tran, A.L. Motivated monitors: The importance of institutional investors' portfolio weights. *J. Financ. Econ.* **2015**, *118*, 21–48. [CrossRef]
- 37. Koh, P.S. Institutional investor type, earnings management and benchmark beaters. *J. Account. Public Policy* **2007**, *26*, 267–299. [CrossRef]
- 38. Bushee, B.J. The influence of institutional investors on myopic R&D investment behavior. *Account. Rev.* **1998**, 305–333.
- Li, Y.; Wang, J.; Wu, X. Distracted institutional shareholders and managerial myopia: Evidence from R&D expenses. *Financ. Res. Lett.* 2019, 29, 30–40.
- 40. Cho, S.J.; Chung, C.Y.; Liu, C. Does Institutional Blockholder Short-Termism Lead to Managerial Myopia? Evidence from Income Smoothing. *Int. Rev. Financ.* **2019**, *19*, 693–703. [CrossRef]
- 41. Freeman, P.; Bloom, M.; Hillman, A.J. Investor activism, executive responsiveness, and corporate social performance. *Strateg. Manag. J.* 2007, *28*, 91–100.
- 42. Lee, J.M.; Park, J.C.; Folta, T.B. CEO career horizon, corporate governance, and real options: The role of economic short-termism. *Strateg. Manag. J.* **2018**, *39*, 2703–2725. [CrossRef]
- 43. Mitchell, R.K.; Agle, B.R.; Wood, D.J. Toward a theory of stakeholder identification and salience: Defining the principle of who and what really counts. *Acad. Manag. Rev.* **1997**, *22*, 853–886. [CrossRef]
- 44. Godfrey, P.C.; Merrill, C.B.; Hansen, J.M. The relationship between corporate social responsibility and shareholder value: An empirical test of the risk management hypothesis. *Strateg. Manag. J.* **2009**, *30*, 425–445. [CrossRef]
- 45. Cao, X.; Lemmon, M.; Pan, X.; Qian, M.; Tian, G. Political promotion, CEO incentives, and the relationship between pay and performance. *Manag. Sci.* **2019**, *65*, 2947–2965. [CrossRef]
- 46. Kasznik, R.; McNichols, M.F. Does meeting earnings expectations matter? Evidence from analyst forecast revisions and share prices. *J. Account. Res.* **2002**, *40*, 727–759. [CrossRef]
- Matsumoto, D.A. Management's incentives to avoid negative earnings surprises. *Account. Rev.* 2002, 77, 483–514. [CrossRef]
- 48. Gaspar, J.M.; Massa, M.; Matos, P. Shareholder investment horizons and the market for corporate control. *J. Financ. Econ.* **2005**, *76*, 135–165. [CrossRef]
- 49. Liu, X. Impression management against early dismissal? CEO succession and corporate social responsibility. *Corp. Soc. Responsib. Environ. Manag.* **2019**. [CrossRef]
- 50. Sanders, W.G. Behavioral responses of CEOs to stock ownership and stock option pay. *Acad. Manag. J.* **2001**, 44, 477–492.

- 51. Chen, Y.F.; Lin, F.L.; Yang, S.Y. Does institutional short-termism matter with executive myopia? *J. Bus. Res.* **2015**, *68*, 845–850. [CrossRef]
- 52. Gloßner, S. Investor horizons, long-term blockholders, and corporate social responsibility. *J. Bank. Financ.* **2019**, *103*, 78–97. [CrossRef]
- 53. Minor, D.; Morgan, J. CSR as reputation insurance: Primum non nocere. *Calif. Manag. Rev.* 2011, 53, 40–59. [CrossRef]
- 54. Xu, L.; Zhang, S.; Liu, N.; Chen, L. Corporate hypocrisy: Role of non-profit corporate foundations in earnings management of for-profit founder firms. *Sustainability* **2018**, *10*, 3991. [CrossRef]
- 55. Dechow, P.M.; Sloan, R.G.; Sweeney, A.P. Detecting earnings management. Account. Rev. 1995, 193–225.
- 56. Jones, J.J. Earnings management during import relief investigations. J. Account. Res. **1991**, 29, 193–228. [CrossRef]
- Arli, D.; Grace, A.; Palmer, J.; Pham, C. Investigating the direct and indirect effects of corporate hypocrisy and perceived corporate reputation on consumers' attitudes toward the company. *J. Retail. Consum. Serv.* 2017, 37, 139–145. [CrossRef]
- 58. Philippe, T.W.; Koehler, J.W. A factor analytical study of perceived organizational hypocrisy. *Sam Adv. Manag. J.* **2005**, *70*, 13.
- 59. Lins, K.V.; Servaes, H.; Tamayo, A. Social capital, trust, and firm performance: The value of corporate social responsibility during the financial crisis. *J. Financ.* **2017**, *72*, 1785–1824. [CrossRef]
- 60. Flammer, C. Corporate social responsibility and shareholder reaction: The environmental awareness of investors. *Acad. Manag. J.* 2013, *56*, 758–781. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).